Author's response to reviews

Title: Trastuzumab with either docetaxel or vinorelbine as first-line treatment for patients with HER2-positive advanced breast cancer: a retrospective comparison

Authors:

Stefania Redana (stefania.redana@gmail.com)
Michela Donadio (mdonadio@molinette.piemonte.it)
Franco Nolè (franco.nole@ieo.it)
Maria Elena Jacomuzzi (elena.jacomuzzi@virgilio.it)
Alessandra Beano (gv.beano@accademiaservice.it)
Rossella Martinello (ross.elisewin@libero.it)
Anna Sapino (anna.sapino@unito.it)
Giuseppe Viale (giuseppe.viale@ieo.it)
Massimo Aglietta (massimo.aglietta@ircc.it)
Filippo Montemurro (filippo.montemurro@ircc.it)

Version: 5 Date: 28 January 2010

Author's response to reviews:

Dear Editor and BMC Cancer, Editorial Production Team, we are submitting a revised version of ms_1525853606286982. In this version we have introduced the requested changes to the format of the manuscript. In particular:

Major revision

1) each of the three figures has now being numbered and uploaded as a separate (figure 1, figure 2 and figure 3). The text and the figure legend have been changed accordingly.

Minor changes:

1) Methods has been renamed methods as requested
2) The vertical lines have been removed by the tables.

Furthermore, some typos have been corrected

We hope that in its current format, our manuscript is suitable for publications.

Yours Sincerely
Filippo Montemurro, MD.